Author's response to reviews

Title: Relationship between home care service use and changes in the care needs level of elderly Japanese

Authors:

Gohei Kato (goheik@saitama-med.ac.jp)
Nanako Tamiya (ntamiya@md.tsukuba.ac.jp)
Masayo Kashiwagi (mkashi@md.tsukuba.ac.jp)
Mikiya Sato (mikiya@sa.to)
Hideto Takahashi (takahasi@md.tsukuba.ac.jp)

Version: 6 Date: 14 December 2009

Author's response to reviews: see over
Dear Dr. Hans Zauner,

I appreciate your help. We add small change to our article. I would like to submit the revised manuscript.

The changes we made are as follows.

1. The title is changed
   From: Relationship between home care service use and changes in the care needs level of elderly Japanese
   To: Relationship between home care service use and changes in the care needs level of Japanese elderly

2. We changed the authors’ personal information.

Sincerely yours,

Gohei KATO